Author's response to reviews

Title: Prevalence and Clinical Impact of Magnesium Disorders in End Stage Renal Disease: A Protocol for a Systematic Review

Authors:

  John Floridis (john.floridis@nt.gov.au)
  Asanga Abeyaratne (Asanga.Abeyaratne@nt.gov.au)
  Sandawana W Majoni (William.Majoni@nt.gov.au)

Version: 4 Date: 4 March 2015

Author's response to reviews: see over
Prevalence and Clinical Impact of Magnesium Disorders in End Stage Renal Disease: A Protocol for a Systematic Review

John Floridis¹, Asanga Abeyaratne¹, Sandawana William Majoni¹, ²

Cover letter addressing previous concerns by the reviewer

Major revisions include:
- We have tried to readdress the original comments made in a previous version of the manuscript, by including much more detail about the quality assessment methods. The Newcastle-Ottawa System (NOS) has been the tool selected which most appropriately suits our systematic review. We have also included an additional file which provides an example of the table we will be using for quality assessment of papers.
- We have removed dichotomous measures, in accordance with our previous intention of changing this manuscript from a meta-analysis to a pure systematic review.
- We have outlined exactly what questions were are seeking from papers, and how this relates to our original title.
- We have revised our approach, from PRISMA to PRISMA-P, in accordance with the advice provided by the previous reviewer. We feel this is more appropriate and are appreciative of the suggestion.

Track changes, and inserted comments have also been incorporated to assist the Editors in reviewing the revised version of the manuscript.